Author’s response to reviews

Title: Genomic profiling in ovarian cancer retreated with platinum based chemotherapy presented homologous recombination deficiency and copy number imbalances of CCNE1 and RB1 genes

Authors:

Alexandre Anastácio da Costa (alexandreandredacosta@gmail.com)
Luisa do Canto (luisaamatos@gmail.com)
Simon Larsen (sjlarsen@imada.sdu.dk)
Adriana Ribeiro (argribeiro@gmail.com)
Carlos Stecca (carlosstecca@gmail.com)
Annabeth Petersen (Annabeth.Hogh.Petersen@rsyd.dk)
Mads Aagaard (Mads.Jorgensen@rsyd.dk)
Louise de Brot (loudebrot@gmail.com)
Jan Baumbach (jan.baumbach@imada.sdu.dk)
Glauco Baiocchi (glbaiocchi@yahoo.com.br)
Maria Achatz (miachatz@gmail.com)
Silvia Rogatto (silvia.rogatto2@gmail.com)

Version: 2 Date: 14 Apr 2019

Author’s response to reviews:

Dear Dr Catherine Rice,

We thank you for your revision of the manuscript. We have done the appropriate changes following your comments as follow:
Editorial Comments:

1 - We removed any files from the file inventory which you do not wish to see published.

2 - We renamed 'Ethics approval' to 'Ethics approval and consent to participate'.

3 - For all research involving human subjects, informed consent to participate in the study should be obtained from participants (or their parent or guardian in the case of children under 16) and a statement to this effect should appear in the ‘Ethics approval and consent to participate’ section of the Declarations including whether the consent was written. When reporting on such studies, individual patient data should not be made available unless consent for publication has also been obtained.

If the need for informed consent has been waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this with details, including the name of the Board or a reference to the relevant legislation in the ‘Ethics approval and consent to participate’ section of the Declarations.

We clear stated in the "Ethics approval and consent to participate" that the need for informed consent has been waived by an IRB.

4 - In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section.

We stated that the funding bodies had no role in any part of the study development.

Sincerely,

Alexandre André B A da Costa